Distinguished Editor and Reviewers,

First and foremost, on behalf of my co-authors, we are very grateful to you for giving us an opportunity to revise our manuscript. We appreciate you very much for your positive and constructive comments and suggestions on our manuscript entitled "Tension-Reduced Closure of Large Abdominal Wall Defect Caused by Shotgun Wound: A Case Report and Literature Review"(ID:77639).

Please accept our sincere thanks for the several opportunities to update our work. We have taken the time to consider the reviewer’s recommendations carefully and have completed improvements to the text. We would love to thank you for allowing us to resubmit a revised copy of the manuscript and we highly appreciate your time and consideration. I hope that the corrections will be satisfactory to you.

[Reviewer1]: thank you for submitting a good case report.

Response: Thank you very much for your consideration. Thank you for the recognition of this study, and on behalf of the entire research staff, I would like to express our most sincere gratitude to you!

[Reviewer2]: Dear editors! The manuscript "Tension-Reduced Closure of Large Abdominal Wall Defect Caused by Shotgun Wound: A Case Report and Literature Review" contains information and treatment tactics that are especially relevant today. Gunshot injuries and subsequent complications require a careful and precision approach. The authors revealed the topic of the manuscript and gave full answers to
the questions asked by other reviewers. The references are relevant and correspond to the manuscript. The scientific style and presentation of the material are made at a fairly high level. Of course, this work will arouse interest in the scientific and practical environment of military field surgeons, and will be in demand by plastic surgeons.

Response: We have studied the reviewers' comments carefully and thank you for the recognition of our work. Thanks again for the hard work of the reviewer! Thanks very much for sharing your thoughts. The large full-thickness abdominal defect caused by a gunshot wound was often associated with infection, multiorgan injuries, and poor patient conditions. This paper reports a case of abdominal wall defect due to a gunshot wound and suggested the possibility of using Top Closure® 1S for temporary LAWD. Thank you for your constructive and professional comments, which have been of great benefit to us. We will continue to delve further in this field and strive to bring more valuable research content.

Most sincerely,

Youbai Chen
MD, PhD, MMS
Associate Professor, Department of Plastic and Reconstructive Surgery, Chinese PLA General Hospital, Beijing, China.

Email: youbaichen@foxmail.com
Yan Han, MD, PhD
Professor, Department of Plastic and Reconstructive Surgery, Chinese PLA General Hospital, Beijing, China.

Email: 13720086335@163.com